Author's response to reviews

Title: Oral maxillofacial neoplasms in an East African population a 10 year retrospective study of 1863 cases using histopathological reports.

Authors:

Adriane Kamulegeya (adrianek55@163.com)
Boniphace Kalyanyama (kalyanyamab@live.com)

Version: 7 Date: 1 July 2008

Author's response to reviews: see over
Qualifications deleted
Names of authors put on same line
Email addresses included as suggested
Title removed from abstract page
Introduction changed to background
Materials and methods changed to Methods
Author contribution corrected to author’s contributions

N.B
The tables have been formatted to simple1 style but cant fit them on a single page. Its something that was raised by the reviewers but given the nature of the study and results there was no way out.
If your software can allow for the tables to fit on a single page we allow you to go ahead and change them.

Figure removed from the main text and attached as a figure. The legend is with in the figure because have used excel for analysis all that was input and cant be deleted from the figure.